%
% main.tex -- Paper zum Thema <reedsolomon>
%
% (c) 2021 Joshua Bär und Michael Steiner, Hochschule Rapperswil
%
\chapter{Reed-Solomon-Code\label{chapter:reedsolomon}}
\lhead{Reed-Solomon-Code}
\begin{refsection}
\chapterauthor{Joshua Bär und Michael Steiner}

% Joshua
\input{papers/reedsolomon/einleitung.tex}
\input{papers/reedsolomon/idee.tex}
\input{papers/reedsolomon/dtf.tex}

% Michael
\input{papers/reedsolomon/endlichekoerper}
\input{papers/reedsolomon/codebsp}
\input{papers/reedsolomon/decohnefehler}
\input{papers/reedsolomon/decmitfehler}
\input{papers/reedsolomon/rekonstruktion}
\input{papers/reedsolomon/zusammenfassung}
\input{papers/reedsolomon/anwendungen}
\input{papers/reedsolomon/hilfstabellen}

\nocite{reedsolomon:weitz}
\nocite{reedsolomon:informationkommunikation}
\nocite{reedsolomon:voyager_programm}
\nocite{reedsolomon:voyager}
\nocite{reedsolomon:cd_wiki}
\nocite{reedsolomon:cd}
\nocite{reedsolomon:qr_wiki}
\nocite{reedsolomon:qr}
%\nocite{reedsolomon:mendezmueller}

\printbibliography[heading=subbibliography]
\end{refsection}
